
CENWP-OP-B                                                                                                                                  23 April, 2002

MEMORANDUM FOR CENWP-OP, ATTN: Gary Johnson

SUBJECT: Bonneville Lock and Dam, Fishway and Fish Activities for week 16 of 2002, which covers the period 14 through 20 April 2002.
1.  OPERATION SUMMARY.

1.1 Daily river flow averaged 252.8 to 348.4kcfs.  Daily project tailwater elevation averaged 20.3’ to 25.7’ msl.  Daily project forebay elevation averaged 75.7’ to 76.5’ msl.  River temperature averaged 53 to 56° F.  Daily turbidity measured 2.0 to 2.8’ on the secchi disk.

1.2 Spill Bays 1and 18, which supply attraction flow for Cascades Island and Bradford Island ladders respectively, are operational and are included in the new spill pattern.  Spill averaged 76.2 to 159.8kcfs for the week.  
1.3 Unit 1 remains out of service for digital governor installation.  Unit 5 remains out of service for turbine rehab.  Unit 6 was out of service from 0930 until 1654 on 16 April for MGR test equipment installation.  Unit 7 returned to service at 1636 on 18 April. Unit 8 was out of service from 1212 until 1330 on 16 April due to an electrical ground.  Unit 17 was out of service for FGE testing from around 1700 until 2400 during the week.    
1.4 Second Powerhouse fish units provide attraction flow for the Washington Shore fish ladder.  The units are placed on standby from approximately 0000-0300 nightly if drawdown exceeds 1.0’.  Fish Unit 1 and 2 were out of service for trashrack raking from 1342 until 1636 on 16 April.
1.5 Gatewell drawdowns were measured on 17 April. 
2.  ACTIVITIES. 
2.1 AWS Closures.  The following exits and AWS valves were closed for cleaning.

15 April- Bradford Island exit from 1545 until 1550, FV5-9 from 1610 until 1620.  

17 April- FV3-7 from 1330 until 1350.

2.2 STS/VBS Inspections.  The following STSs/VBSs were inspected.

	Unit
	Date
	Run Hours
	Comments

	11
	04/10
	750
	Motor failure in C-slot.  Replaced and put back in service.

	12
	04/10
	752
	

	13
	04/15
	864
	

	14
	04/15
	829
	

	15
	04/08
	587
	Screens moved to Unit 17 for FGE test

	16
	04/17
	890
	

	17
	04/08
	42
	Screens installed from Unit 15

	18
	04/17
	919
	


Table 1.  STS Inspection Dates.

2.3 Dewatering and Fish Salvage.  DSM1 was turned south at 0715 on 15 April for repairs to the small trash sweep.  2 Spring Chinook adults, 4 steelhead smolts and 2 chinook smolts were removed and released.

3.  RESEARCH.

3.1 CRITFC Age and Length Composition. Collected 90 Spring Chinook.  They will be collecting Chinook and Coho twice a week from 1 April to 31 October.

3.2 NMFS FGE.  Unit 17 FGE study is underway.  

3.3 NMFS Lamprey.  Tracking continues.   Collection begins in June.

3.4 NMFS Saltwater Rearing.  Collection has begun.  Fish will be held until December.

3.5 Normandeau Balloon-tag at Spillway.  Scheduled for 12-31 May and 1-15 August.

3.6 OCFWRU Transportation Study.  Study of post-transportation migration and survival of juveniles to run 29 April through 14 July.

3.7 ODFW Summer Steelhead sampling.  Sampling 2 days/week 8 hours/day.  Will sample 1% of the run.

3.8 ODFW Pikeminnow Electroshocking.  Part of northern pikeminnow management program.  Capturing and collecting biological data from pikeminnow, smallmouth bass, and walleye in Bonneville’s BRZ.  Electroshocking  1-5 April and 22-26 April.

3.9 Pelfrey Sturgeon Hatchery.  Scheduled to begin capturing broodstock in late May.

3.10 PSMFC Adult PIT.  Detectors are installed in all ladders.  Some of the detectors have developed leaks and will be monitored throughout the season.  Repairs are scheduled for 2002-03 winter maintenance season.

3.11 PSMFC JMF.  Debris at the primary dewatering structure (PDS) and fish/debris separator was heavy during the week.  Water flow across the PDS was stable throughout the week.  The screen cleaner cycled 4 times per hour.  The following fish were found stranded on the separator bars: 4 smolts, 9 kelts (released alive) and 7 kelts arrived dead.  Fish recorded crossing the separator bars included: 299 steelhead, 35 salmon, 37 unknown salmonids and 366 smolts.

3.12 PSMFC/NMFS Juvenile PIT.  See Misc. 5.3.
3.13 U of I Transportation Study.  Evaluation of physiological condition of river run v. barged Snake River juveniles to occur May-June.  May add separation–by-code study June-August.

3.14 U of I Adult Radio tracking.  Trapping began 1 April and will continue 7 days/week until October.

3.15 USGS Juvenile Radio tracking.  Study in progress.

3.16 USGS MGR.  Equipment installation continues.  

3.17 USGS EMG studies.  Tissue samples will be taken from 200 Chinook starting 1 April through June.

3.18 USGS Lamprey.  Collection begins in May.

3.19 WDFW Fish Counting.  16-hour visual and 8-hour video counts continue at Bradford Island and Washington Shore ladders.

3.20 WDFW Jaw tagging.  Tagging 50-75 fish per week, starting 1 April.

3.21 WDFW Winter Steelhead Tagging.  Trapping and tagging of steelhead continues one day a week.

3.22 WES Spillway Deflector Testing.  WES will be monitoring water quality equipment to determine effectiveness of new flip lips.  Study began with initiation of spill on 10 April.

3.23 WES/Battelle/Mevatech Juvenile Hydroacoustics.  Study in progress.  

4.  FISHWAYS 
4.1 Project biologists inspected 14-16, 18 April this reporting week.

4.2 Fish Passage Plan Observations.

	DATE
	LOCATion
	FPP Deviation
	CAUSE
	RESPONSE
	TR/WAR

	4/14-16
	PH1 PCC velocity
	Velocity <1.5fps on the electronic meter.
	Unknown- the rest of the ladder is in criteria.
	No action at this time.
	

	4/17, 4/19-20
	
	Biologists did not inspect 7 days/ week.
	Personnel shortage.
	Job advertised.
	

	4/14
	FV3-7
	Not controlling properly.
	Unknown
	Operators notified.  Electricians to repair.
	

	4/14
	PH2 PCC velocity
	Velocity >1.5fps on the electronic meter.
	Unknown- the rest of the ladder is in criteria.
	No action at this time.
	

	4/15-16
	DSM1
	Out of service
	Small trash sweep broken.
	DSM turned south.  

See misc. 5.3
	

	4/16
	Cascade Island
	Diffusers not in proper position for TW elevation.
	Computer system not yet fully functioning
	Operators informed, adjusted diffusers.
	

	4/16-18
	DSM2
	Screen sweep #2 not functioning.  Add in water valve closed.
	Broken chain, sprockets, motor.  Water levels high.
	 #1 and  #3 in continuous mode.   See misc. 5.4
	

	4/18
	Weir 37
	Reading > or < 1.0’
	Stillwell needed.
	Added to winter maint. list
	

	4/18
	Orifice 17 B-N
	Closed
	Was forgotten after FGE testing.
	Operators opened the orifice.
	


4.3 Fish Ladder Temperatures.  Temperatures are measured 3 weirs down from the count station at Bradford Island and at the junction of the UMT and the main ladder at Washington shore.   The temperature is recorded every 15 seconds, averaged for each hour then averaged for the entire day.  Temperatures are in Fahrenheit.

	Date
	Wash. Shore
	Bradford Is.

	4/14
	49.1
	49.1

	4/15
	49.1
	49.1

	4/16
	49.1
	49.1

	4/17
	49.5
	49.3

	4/18
	49.8
	49.8

	4/19
	50
	50

	4/20
	50
	50


4.4 Sensor Calibration.  Calibration was checked on 16 April this reporting week.  Everything is within criteria.

4.5 Zebra Mussel Monitoring.  No signs of colonization were observed this reporting week. 

5. MISCELLANEOUS

5.1 FV4-3, FV4-4, FV5-3, and FV5-4 are operated in manual until the computer system in up and running.  Calibration should be completed sometime in April.

5.2 River temperature is the temperature of the water in the scroll case of Unit 2.  This temperature is recorded every 6 hours (the FPP requires a reading three times a day). 

5.3 Sampling and flat plate operations were interrupted from 1330 on Sunday 14 April to 1030 on Tuesday, 16 April to repair the small dewatering screen trash sweep.  On Wednesday, 17 April at 2130 damage to the Flat Plate and small dewatering screen halted operations again.   Repairs are scheduled for Tuesday, 23 April when replacement parts arrive.

5.4 At 1950 on 14 April, screen sweep #2 at DSM2 failed.  New chain was installed at that time.  At 0440 on 17 April, screen sweep #2 failed again.  Due to high debris load, sweeps #1 and #3 were operated in continuous manual mode.  The add-in water valve was closed to prevent high water levels from tripping the ERG while screen sweep #2 was OOS.  Everything was back in working order by 1655 on 18 April.

5.5 FV3-7 went to manual operation at 1600 on 16 April due to a failure to control properly.  Electricians corrected the problem and returned the valve to auto at 1000 on 18 April.

5.6 PH1 ITS was closed from 0815 until 0830 on 17 April for project work.

5.7 Sea lions and harbor seals have been observed in the tailrace of both powerhouses and the spillway.  No action is being taken by the project to harass, harm, or otherwise annoy these animals.  FFU is monitoring and recording their activity.

5.8 Turbulent flows from spillbays 1 and 18 enter the side entrance gates of Cascade Island and Bradfrod Island B-branch ladders, causing poor entrance conditions.  District hydraulic engineers are investigating the problem.  BON proposes bulkheading the side entrances.  

 DEBORAH L. CHENOWETH

                                 

                                                             Operations Project Manager
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